Author’s response to reviews

Title: Home management of lower limb lymphoedema with an intermittent pneumatic compression device: A Feasibility Study

Authors:

Nyree Dunn (nyree.dunn@southwales.ac.uk)

E Williams (mark.williams@southwales.ac.uk)

Michelle Fishbourne (michelle.fishbourne@wales.nhs.uk)

Gina Dolan (gina.dolan@southwales.ac.uk)

Jane Davies (daviesj112@cardiff.ac.uk)

Version: 4 Date: 03 Sep 2019

Author’s response to reviews:

Dear Editor

Subject: Submission of revised paper PAFS-D-18-00147

Home management of lower limb lymphoedema with an intermittent pneumatic compression device: A Feasibility Study – Pilot and Feasibility Studies

We have carefully reviewed the comments and have revised the manuscript accordingly. Our responses are given in a point-by-point manner below. Changes to the manuscript can be found using the page and line references below. We hope the revised version is now suitable for publication and look forward to hearing from you in due course.

Sincerely,

Nyree Dunn

University of South Wales

2) The checklist has been changed to the CONSORT extension to pilot trials